Author’s response to reviews

Title: ACR TI-RADS and ATA US scores are helpful for the management of thyroid nodules with indeterminate cytology

Authors:

Thayse Madsen Barbosa (thayselozovoy@hotmail.com)
Cleo Mesa Junior (cleomesa@uol.com.br)
Hans Graf (hansgraf@bighost.com.br)
Teresa Cavalcanti (tecava@yahoo.com.br)
Marcus Trippia (trippia.s@terra.com.br)
Ricardo da Silveira Ugino (ricardougino@gmail.com)
Gabriel Lucca de Oliveira (glucca11@gmail.com)
Victor Granella (victor_granella@hotmail.com)
Gisah Amaral de Carvalho (carvalho.gisah@gmail.com)

Version: 2 Date: 08 Sep 2019

Author’s response to reviews:

"ACR TI-RADS and ATA US scores are helpful for the management of thyroid nodules with indeterminate cytology"

(BEND-D-19-00264R1)

Response Letter

Editor Comments:

1. The corresponding author email address is different on the Editorial Manager system to in the manuscript. Please change this so that these email addresses match.

The e-mail address was changed accordingly to the Editorial Manager system.

2. Please rename the Introduction section to Background.
It was renamed Introduction section to Background.

3. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

It was added “Conclusions” section.

4. Please rename the Materials and methods section to Methods.

It was renamed Materials and methods section to Methods.

5. As many abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.

All abbreviations are discriminated before the declaration section.

6. Before the Author disclosure statement, please add a Declarations heading.

It was added a Declarations heading.

7. In the Declarations section, please add an Ethics approval and consent to participate section.

It was added an Ethics approval and consent to participate section.

8. Please include a statement on ethics approval and consent to participate in the “Ethics and consent to participate” section of the Declarations. Please include the full name of the ethics committee (and the institute to which it belongs) that approved the study and the committee’s reference number if appropriate. If you did not need formal ethics approval please confirm that this complies with national guidelines and provide a reference which supports this. Alternatively, supply a statement that says that a local ethics committee ruled that no formal ethics approval was required in this particular case.
It was provided the statement on ethics approval and consent, including the full name of the ethics committee that approved the study and the committee’s reference number (CAAE: 45277115.3.0000.0096).

9. We note that you have not included a ‘Consent for publication’ section in the Declarations. If identifying images or other personal or clinical details of participants are presented that compromise anonymity, a statement of consent to publish from the patient should be included. This section must be included even if it is not applicable to your manuscript. If consent to publish is not applicable to your manuscript please write ‘Not Applicable’ in this section.

It was added a “Consent for publication” in the Declarations section.

10. We note that you have not included an ‘Availability of data and materials’ section in the Declarations. Please add this section.

It was included an ‘Availability of data and materials’ section.

11. In the Declarations section, please add an 'Author's contributions' section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section.

It was added an ‘Author’s contributions’ section.

12. Please rename the Author Disclosure Statement section to Competing interests.

It was renamed Author Disclosure Statement section to Competing interests.

13. Please remove the additional files attached to the manuscript, as they are no longer required.

The additional files attached to the manuscript was removed and it was uploaded as separate files.

14. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.
All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The manuscript and relevant tables/figures/additional files were uploaded and are in the file inventory.